Author's response to reviews

Title: Preparedness of Emergency Departments in NorthWest England for Managing Chemical Incidents: a structured interview survey

Authors:

Jane Williams (janelwilliams@doctors.org.uk)
Darren Walter (darren.walter@smtr.nhs.uk)
Kirsty Challen (kirsty.challen@smtr.nhs.uk)

Version: 2 Date: 21 June 2007

Author's response to reviews: see over
Editorial
A competing interests section has been inserted.

Reviewer 1:
Grammatical corrections have been made.

Reviewer 2:
No alterations.

Reviewer 3:
The discussion section regarding responsibility for contaminated water has been clarified. We have added some comments regarding capacity management.